

Kinesiology (Kinesio) Taping

Optum Health Solutions Musculoskeletal (MSK) Utilization Management Policy Policy Number: 483

Effective Date: 04/24/2025

Optum is a registered trademark of Optum, Inc. in the U.S. and other jurisdictions. All other brand or product names are the property of their respective owners. Because we are continuously improving our products and services, Optum reserves the right to change specifications without prior notice. Optum is an equal opportunity employer.

Table of Contents

Policy Statement	3
Purpose	3
Summary	3
Scope	3
Description	3
Clinical Evidence	3
Conclusion	6
References	7
Review and Approval History	9
Plain Language Summary	

Policy Statement

Optum considers Kinesiology (kinesio) taping (KT) therapy unproven and not medically necessary for the treatment of neuromuscular disorders due to insufficient scientific evidence of the effectiveness as either a single intervention or when combined with other treatment.

Purpose

This policy describes the position of Optum regarding the efficacy, effectiveness, risks, and burdens associated with the use of KT.

Summary

- KT is a therapeutic taping method that utilizes a latex-free elastic tape, which is proposed to give support and stability to joints and muscles without affecting circulation, range of motion (ROM), and biomechanics. It is also used for preventative maintenance, edema, and to treat pain.
- KT is promoted by healthcare practitioners and consumers as having therapeutic effects for the treatment of different musculoskeletal disorders.
- KT is frequently viewed as an adjunct to therapy and exercise.
- Evidence syntheses for a range of musculoskeletal disorders have generally reported favorable trends; however, the clinical relevance of KT on patient important outcomes (pain, function) remains to be established.
- Further research is likely to have an important impact on confidence in the estimate of effect (Basset et al., 2010).

Scope

The application of this policy is limited to those services and supplies best described as KT. Conventional athletic taping and McConnell taping are excluded from the scope of this policy.

Description

The KT method is applied over muscles to reduce pain and inflammation, relax overused and tired muscles, and to support muscles in movement on a 24hr/day basis. It is a non-restrictive type of taping, which allows for full ROM.

Clinical Evidence

Different from traditional rigid tape, KT is an elastic woven-cotton strip that can be stretched from 120% to 140% of its original length and can be kept in situ for 1-5 days at a time. KT purportedly mimics the physical qualities of the skin as it is believed to be the same weight and thickness of the epidermis along with its inherent elastic properties. There are many proposed benefits to KT including proprioceptive facilitation, reduced muscle fatigue, muscle facilitation, reduced delayed-onset muscle soreness, pain inhibition, enhanced healing such as reducing edema, and improvement of lymphatic drainage and blood flow. There has been an increasing number of studies investigating the role of KT for the treatment of pain and disability related to musculoskeletal disorders (MSD) affecting axial and extremity regions (Williams et al., 2012; Basset et al., 2010).

General Musculoskeletal Disorders and Myofascial Pain Syndrome

Banerjee et al. (2019) studied the effects of KT on experimentally-induced pain in otherwise pain-free, healthy participants. The authors conducted a randomized controlled repeated-measures laboratory study. The sample size was 48 participants. The study findings suggest there was no difference in effects when comparing KT, standard taping, and sham taping in response to experimental stimulation in otherwise healthy and pain free participants.

Zhang et al. (2019) conducted a systematic review with meta-analysis that evaluated the efficacy of KT for persons diagnosed with myofascial pain syndrome or having myofascial trigger points. The analysis found there was no detectable effect of KT on disability or function. An appraisal of results showed KT achieved statistically significant

improvement to comparators in pain intensity measurement at post-intervention; however, the results did not achieve clinical relevance.

Head and Face Complaints

The efficacy of KT was judged to be unclear (equivocal), based on moderate quality evidence from four clinical trials, for the treatment of temporomandibular joint dysfunction (Cupler et al., 2020).

Upper Extremity (UE) Conditions

Cupler et al. (2020) assessed the evidence for the effectiveness of KT in the treatment of multiple UE conditions. Based on moderate quality evidence, KT was deemed to show an unclear trend in outcomes for subacromial impingement syndrome, lateral epicondylalgia, carpal tunnel syndrome, and osteoarthritis of the proximal interphalangeal joint. KT was considered promising for the treatment of de Quervain's Syndrome; however, this conclusion was based on weak evidence and is likely to be impacted by future higher quality studies.

Ghozy et al. (2020) conducted a systematic review and meta-analysis that investigated the efficacy of KT as a standalone treatment, as an adjunct treatment to exercise, and compared with other usual treatment modalities for shoulder pain. The authors found insufficient evidence to support the use of KT in clinical practice as a treatment for shoulder pain. There was limited evidence of its benefit as a complement to exercise in the treatment of shoulder pain syndromes; however, the clinical relevance of benefit could not be ascertained.

A systematic review and meta-analysis by Celik et al. (2020) evaluated the effects of KT on shoulder disorders (impingement, nonspecific pain, calcific tendonitis), as a single treatment modality or in conjunction to other treatments. Despite reported positive effects in some studies, the authors found no firm evidence of any benefit of KT on shoulder disorders. Fourteen studies were included with 680 participants. KT did not produce better results on pain compared to sham, exercises, or passive treatments. Similarly, KT was not superior to sham taping, exercises, or passive treatments on function. There were no significant between-group differences with ROM compared to sham taping and passive treatment.

Deng et al. (2021) investigated the effectiveness of KT compared to inert controls (no treatment, sham taping) for the management of hemiplegic shoulder pain. The results of the systematic review and meta-analysis showed a trend favoring KT for pain, upper limb motor function, and the magnitude of shoulder subluxation. Pain outcomes did not reach a clinically relevant effect. The clinical relevance of other outcomes was not described. Passive controls demonstrated a greater effect on activities of daily living when compared to KT intervention.

A meta-analysis from randomized controlled trials (RCTs) was performed by Zhong et al. (2020) in order to evaluate the efficacy and safety of KT in the treatment of lateral epicondylitis. Five studies with a total of 168 patients were included. The meta-analysis described statistically but not clinically superior pain scores, grip strength, Modified Mayo performance index, and disabilities of arm, shoulder, and hand (DASH) functional scores.

Spinal and Pelvic Disorders

The results of RCTs on the effectiveness of KT for chronic nonspecific low back pain (LBP) were summarized in a meta-analysis by Lin et al. (2020). The authors concluded there is low-quality evidence that KT has a beneficial role in pain reduction and disability improvement for patients with chronic nonspecific LBP. More high-quality studies are required to confirm the effects of KT on chronic nonspecific LBP. However, the pooled analyses did not demonstrate clinically meaningful benefits for pain and disability.

An RCT performed by Uzunkulaoğlu et al. (2018) studied the effectiveness of KT on pain in chronic non-specific LBP. Sixty patients were placed in intervention or placebo groups. Intervention group (n=30) were treated with KT. Placebo group (n=30) were treated with sham taping. Patients were evaluated according to pain level, modified Schober's test value, hand-ground distance, and lumbar ROM and disability at baseline at the end of the first and six months of intervention. The authors found KT provided improvements in pain, ROM, and disability in chronic non-specific LBP in the short term. Results may vary due to application methods and frequency. Larger studies with greater patient populations are needed to explain the method for application.

Junior et al. (2019) investigated the effects of KT in patients with nonspecific LBP. Eleven RCTs were included in the systematic review and meta-analysis. Two clinical trials (pooled n=100) compared KT to no intervention at the short-term follow-up. Four studies compared KT to placebo (pooled n=287) at short-term follow-up and two trials (pooled n=100) compared KT to placebo at intermediate-term follow-up. Five trials (pooled n=296) compared KT combined with exercises or electrotherapy to exercises or spinal manipulation alone. No statistically significant difference was found for most comparisons. The authors concluded they found no evidence to support the use of KT in clinical practice for patients with chronic nonspecific LBP.

A systematic review and meta-analysis by Sheng et al. (2019) included eight moderate quality studies in a systematic review and meta-analysis that evaluated the effectiveness of KT, when compared to other non-pharmacologic interventions (physical agents, physical therapy, acupuncture), for patients with chronic non-specific LBP. The pooled effects did not show any clinically relevant between-group differences in pain or disability outcomes.

Lower Extremity Conditions

A mapping review by Cupler et al. (2020) interpreted the quality and direction of the evidence when making recommendations concerning support for the use of KT across a range of lower extremity disorders. Moderate evidence supported a favorable recommendation (a positive trend for KT needs to be confirmed by higher quality RCTs). Promising recommendations, based on weak evidence, were made for the application of KT for tibial stress syndrome and plantar heel pain (plantar fasciitis). For these conditions, the positive but inconclusive results are likely to be impacted by future higher quality studies. The utility of KT for patellofemoral pain syndrome was rated as equivocal, based on moderate evidence. The use of KT for the treatment of ankle sprain yielded an unfavorable recommendation (moderate quality evidence).

Lin et al. (2020) conducted a systematic review and meta-analysis that compared the therapeutic effects between physical therapy (PT) combined with KT and PT alone in knee osteoarthritis treatment. Compared with PT alone, PT combined with KT provided better therapeutic effects regarding pain reduction and functional improvement in patients with knee osteoarthritis. The additional pain reduction and functional improvement did not, however, achieve an important clinical difference. In a systematic review, Melese et al. (2020) sought to summarize the current best evidence for the effectiveness of KT in reducing pain and increasing knee function for patients with knee osteoarthritis. While the findings of individual studies were mixed, in aggregate this qualitative review suggests a favorable trend with KT with improved outcomes for pain and functional disability. This conclusion remains to be confirmed by systematic quantitative analyses. An earlier systematic review and meta-analysis by Ye et al. (2020) reported that KT had significant effects on pain, physical function, ROM, and quadriceps muscle strength in patients with knee osteoarthritis. However, the evidence was insufficient to draw conclusions about clinical relevance and the durability of effects. Wu et al. (2022) performed a systematic review and meta-analysis of RCTs in patients with knee osteoarthritis. A total of 642 patients from 16 RCTs were included in the meta-analysis to evaluate KT plus exercise versus exercise only. The authors concluded that KT plus exercise reduced pain when compared to exercise alone; however, it did not improve knee function. Further high-quality research is needed to evaluate efficacy of KT as a treatment for knee osteoarthritis.

Nunes et al. (2021) investigated whether KT, applied to ankles of healthy people as a preventive intervention and people with ankle injuries (e.g., sprain, instability, tendinopathy), is superior to sham or alternative interventions on ankle function. Eighty-four trials met the eligibility criteria, which evaluated 2,684 people. The systematic reviewers found the current evidence does not support or encourage the use of KT applied to the ankle for improvements in functional performance, regardless the population. The authors concluded that KT technique alone may not be an adequate therapeutic technique to enhance ankle function; and therefore, clinicians should consider techniques with consistent evidence, such as exercise and bracing, to improve postural control, gesture performance, movement, and neuromuscular control, which are related to ankle function. In contrast to Nunes et al. (2021), a previous systematic review and meta-analysis by Wang et al. (2018) reported that KT was superior to other taping methods (athletic taping) in ankle functional performance improvement. An appraisal of this study by Nunes et al. (2019) identified critical flaws including missing studies, suboptimal methodological approaches for assessing the quality of trials, and the analytic approach.

Conclusion

There is insufficient evidence of the efficacy of KT-alone or in combination with other interventions for the treatment of neuromuscular disorders.

References

Banerjee G, Briggs M, Johnson M. The effects of kinesiology taping on experimentally-induced thermal and mechanical pain in otherwise pain-free healthy humans: A randomised controlled repeated-measures laboratory study. PLoS One. 2019 Dec 10;14(12):e0226109.

Basset K, Lingman S, Ellis R. The use and treatment efficacy kinaesthetic taping for musculoskeletal conditions: A systematic review. New Zealand Journal of Physiotherapy 2010;38(2):56-62.

Celik D, Karaborklu A, Coban O, et al. The clinical efficacy of kinesio taping in shoulder disorders: A systematic review and meta-analysis. Clinical Rehabilitation. 2020 Jun;34(6):723-40.

Clinical trials registry. U.S. National Institutes of Health. Accessed:10/18/2024. Available from: https://clinicaltrials.gov/ct2/search

Cupler Z, Alrwaily M, Polakowski E, et al. Taping for conditions of the musculoskeletal system: An evidence map review. Chiropractic & Manual Therapies. 2020 Dec;28(1):1-21.

Deng P, Zhao Z, Zhang S, et al. Effect of kinesio taping on hemiplegic shoulder pain: A systematic review and metaanalysis of randomized controlled trials. Clinical Rehabilitation. 2021(in press);16:0269215520964950.

Ghozy S, Dung N, Morra M, et al. Efficacy of kinesio taping in treatment of shoulder pain and disability: A systematic review and meta-analysis of randomised controlled trials. Physiotherapy. 2020 Jun 1;107:176-88.

Luz Júnior M, Almeida M, Santos R, et al. Effectiveness of kinesio taping in patients with chronic nonspecific low back pain: A systematic review with meta-analysis. Spine (Phila Pa 1976). 2019 Jan 1;44(1):68-78.

Lin C, Lee M, Lu K, et al. Comparative effects of combined physical therapy with Kinesio taping and physical therapy in patients with knee osteoarthritis: A systematic review and meta-analysis. Clinical Rehabilitation. 2020 Aug;34(8):1014-27.

Lin S, Zhu B, Huang G, et al. Short-term effect of Kinesio taping on chronic nonspecific low back pain and disability: A meta-analysis of randomized controlled trials. Physical Therapy 2020;100(2):238-254.

Melese H, Alamer A, Temesgen M, et al. Effectiveness of Kinesio taping on the management of knee osteoarthritis: A systematic review of randomized controlled trials. Journal of Pain Research. 2020;13:1267.

Nunes G, de Noronha M. Do we really have evidence that Kinesio taping improves ankle functional performance? Clinical Rehabilitation. 2019 Mar;33(3):584-6.

Nunes G, Feldkircher J, Tessarin B, et al. Kinesio taping does not improve ankle functional or performance in people with or without ankle injuries: Systematic review and meta-analysis. Clinical Rehabilitation. 2021;35(2):182–199.

Sheng Y, Duan Z, Qu Q, et al. Kinesio taping in treatment of chronic non-specific low back pain: A systematic review and meta-analysis. Journal of rehabilitation medicine. 2019 Nov 5;51(10):734-40.

Uzunkulaoğlu A, Aitken M, Ay S, et al. The effectiveness of Kinesio taping on pain and clinical features in chronic nonspecific low back pain: A randomized controlled clinical trial. Turkish Journal of Physical Medicine and Rehabilitation. 2018 Jun;64(2):126.

Wang Y, Gu Y, Chen J, et al. Kinesio taping is superior to other taping methods in ankle functional performance improvement: A systematic review and meta-analysis. Clinical Rehabilitation. 2018 Nov;32(11):1472-81.

Williams S, Whatman C, Hume P, et al. Kinesio taping in treatment and prevention of sports injuries: A meta-analysis of the evidence for its effectiveness. Sports Med.2012;42:153–164.

Wu H, Yao R, Wu J, et al. Does kinesio taping plus exercise improve pain and function in patients with knee osteoarthritis?: A systematic review and meta-analysis of randomized controlled trials. Front Physiol. 2022 Sep 9;13:961264.

Ye W, Jia C, Jiang J, et al. Effectiveness of elastic taping in patients with knee osteoarthritis: A systematic review and meta-analysis. American Journal of Physical Medicine & Rehabilitation. 2020 Jun 1;99(6):495-503.

Zhang X, Liu L, Wang B, et al. Evidence for kinesio taping in management of myofascial pain syndrome: A systematic review and meta-analysis. Clinical Rehabilitation 2019 Feb 4:269215519826267.

Zhong Y, Zheng C, Zheng J, et al. Kinesio tape reduces pain in patients with lateral epicondylitis: A meta-analysis of randomized controlled trials. International Journal of Surgery. 2020 Apr 1;76:190-9.

Review and Approval History

Date	Action
07/15/2010	Original effective date
10/26/2010	Policy rebranded to "OptumHealth Care Solutions, Inc. (OptumHealth)"
04/07/2011	Annual review and approval completed
04/19/2012	Annual review and approval completed
04/18/2013	Annual review and approval completed
04/17/2014	Annual review and approval completed; Policy rebranded "Optum* by OptumHealth Care Solutions, Inc."
04/16/2015	Annual review and approval completed
04/21/2016	Literature review revised. Coding section deleted. Annual review and approval completed
04/20/2017	Annual review and approval completed; Legal entity name changed from "OptumHealth Care Solutions, Inc." to "OptumHealth Care Solutions, LLC."
04/26/2018	Annual review and approval completed; updated references
04/25/2019	Annual review and approval completed; Revised the Literature Review; Deleted Table 1; Updated references
04/23/2020	Annual review and approval completed; The Literature Review and References sections were updated
04/22/2021	Annual review and approval completed; The Background, Literature Review, Summary, and References sections were updated
05/3/2022	Annual review and approval completed
06/29/2022	Updated legal entity name "OptumHealth Care Solutions, LLC." to *Optum™ Physical Health ("Optum") includes OptumHealth Care Solutions, LLC; ACN Group IPA of New York, Inc.; ACN Group IPA of California, Inc. d/b/a OptumHealth Physical Health of California; Managed Physical Network, Inc.; and OrthoNet Holdings, Inc. which includes OrthoNet New York IPA, Inc., OrthoNet West, Inc., OrthoNet, LLC, OrthoNet of the South, Inc.
04/27/2023	Annual review and approval completed; no significant changes made to the document. Updated contact email from policy.inquiry@optumhealth.com to phpolicy_inquiry@optum.com .
02/14/2024	Annual review completed, no substantive changes to clinical content. Document content transitioned to new policy template. Approved by Optum Clinical Guideline Advisory Committee.
04/25/2024	Annual review and approval completed by Optum Quality Improvement Committee.
12/11/2024	Annual review completed with no substantive changes. Approved by Optum Clinical Guideline Advisory Committee.
04/24/2025	Approved by Optum Quality Improvement Committee.

Plain Language Summary

Kinesiology (Kinesio) Taping (KT)

Utilization Management Policy # 483

Plain Language Summaries are provided by Optum to supplement the associated clinical policy or guideline. These summaries are not a substitute for advice from your own healthcare provider.

What is KT and what is known about it so far?

KT is a thin, stretchy, and hypoallergenic tape. It has been used for both spinal and extremity conditions. KT is used for pain relief, to decrease swelling and inflammation, and support overused muscles.

Information about KT is easily found on vendor and healthcare websites. The uses of KT are largely based on laboratory studies performed on healthy individuals and low-quality clinical research. There is a lack of higher quality information, which is usually needed to make confident judgments about benefits and risks.

How was KT evaluated?

A work group of clinicians was assigned to review the available research. The internet was searched for articles about KT. The work group independently examined the selected research studies. A broadly accepted rating scale was used. Possible ratings were high, moderate, low, or very low quality. Additionally, the positions and guidelines of other professional and healthcare groups were evaluated.

Before it was approved, the policy was presented to a series of committees that included independent health care practitioners.

What did the work group find?

There is only limited research about the effectiveness of KT for the treatment of spinal and extremity disorders. The overall research quality was rated as low. Better quality studies are needed.

It was not possible to decide that KT provided more benefit or less risk, when compared to generally accepted and safe treatments including traditional taping procedures.

What are the limitations of the information?

Several studies involve only healthy people. Others include very specific groups, such as only women. So, it is not clear if positive results apply to different groups.

The use of KT for many spinal and extremity disorders has not been studied.

What are the conclusions?

KT is viewed as unproven and not medically necessary. Further research is needed before its use can be considered an established treatment option for any spinal or extremity condition.